Author's response to reviews

Title: Choroid metastases revealing primary clear cell adenocarcinoma of lung rewarding result by cisplatin and pemetrexed: a case report

Authors:

Chang Yueh shih (yshichang@gmail.com)
Yeung Ling (ing@cgmh.org.tw)
Chang Liang-Che (ing@cgmh.org.tw)
Huang Jen-Seng (ing@cgmh.org.tw)
Yeh Kun-Yun (yehtyng@gmail.com)

Version: 2 Date: 1 March 2013

Author's response to reviews:

Dear Editors:

Thanks for your suggestion. We had corrected our previous article according to the guideline for author. If there is still something need to revise, please let us know and we would complete it as soon as possible. Thanks!

Sincerely, Dr. Chang, Yueh-Shih